Author's response to reviews

Title: Urethral caruncle in a 9-year-old female: A rare case report and review of the literature

Authors:

Masahiro Chiba Dr. (mc-psurg@med.showa-u.ac.jp)
Akira Toki Dr. (atoki@med.showa-u.ac.jp)
Akihide Sugiyama Dr. (akihide@mte.biglobe.ne.jp)
Rie Suganuma Dr. (watanaberiejp@yahoo.co.jp)
Shunsuke Osawa Dr. (colorful_tone@yahoo.co.jp)
Rie Ishii Dr. (reasonal36picture@gmail.com)
Tomokazu Nakagami Dr. (gamigami0123@coast.ocn.ne.jp)
Junichi Suzuki Dr. (mdjun@med.showa-u.ac.jp)
Yu Watarai Dr. (yuwatarai@hotmail.co.jp)
Shinya Kawano Dr. (s_kawano@s01.itscom.net)
Koumei Suzuki Dr. (sukome@ezweb.ne.jp)

Version: 2  Date: 7 November 2014

Author's response to reviews: see over
November 5, 2014

Editorial team
Journal of Medical Case Reports

My co-authors and I would like to submit the attached manuscript titled “Urethral caruncle in a 9-year-old girl: A rare case report and review of the literature” for publication in Journal of Medical Case Reports as a Brief Communication.

This report highlights urethral caruncle, which is of special interest because it is very rare at this age and this is the first case that was cured by conservative treatment. The patient has provided permission to publish these features of his/her case, and the identity of the patient has been protected.

Because your journal welcomes papers dealing with clinical surgery in children, we believe that the findings of this report are relevant to the scope of your journal and will be of interest to its readership.

This manuscript has not been published or presented elsewhere in part or in entirety, and is not under consideration by any other journal. All study participants provided informed consent, and the study design was approved by the appropriate ethics review boards. All the authors have approved the manuscript and have agreed with its submission to your esteemed journal. There are no conflicts of interest to declare.

Thank you for your consideration. I look forward to hearing from you.

Masahiro Chiba
Division of Pediatric Surgery
Department of Surgery
Showa University Hospital
1-5-8 Hatanodai, Shinagawa-ku
Tokyo 142-8666, Japan
Tel: +813-3784-8000
Fax: +813-3784-8789
E-Mail: mc-psurg@med.showa-u.ac.jp